

JGR Atmospheres

RESEARCH ARTICLE

10.1029/2022JD037915

Key Points:

- CH₄ emission factor for coal mining was 23.2(±4.9) m³ CH₄/ton coal, close to the default value of high CH₄-content coal in China
- The *posteriori* CO₂ emissions were 1.6-fold of the a priori emissions, indicating CO₂ emissions in industrial cities were largely underestimated
- Some large CO₂ emissions are missing and more work is needed for urban government to fully understand their greenhouse gases emissions

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

C. Hu, T. J. Griffis, Q. Xiao and L. Zhu, nihaohucheng@163.com; tgriffis@umn.edu; qtxiao@niglas.ac.cn; zhlyun@126.com

Citation:

Hu, C., Xiao, W., Griffis, T. J., Xiao, Q., Wang, S., Zhang, Y., et al. (2023). Estimation of anthropogenic CH₄ and CO₂ emissions in Taiyuan-Jinzhong region: One of the world's largest emission hotspots. *Journal of Geophysical Research: Atmospheres*, *128*, e20221D037915. https://doi. org/10.1029/2022JD037915

Received 24 SEP 2022 Accepted 28 MAR 2023

Author Contributions:

Conceptualization: Cheng Hu Funding acquisition: Cheng Hu Investigation: Cheng Hu Methodology: Cheng Hu Resources: Cheng Hu, Timothy J. Griffis, Qitao Xiao, Lingyun Zhu Supervision: Cheng Hu Validation: Cheng Hu Validation: Cheng Hu Writing – original draft: Cheng Hu Writing – review & editing: Cheng Hu

© 2023. American Geophysical Union. All Rights Reserved.

HU ET AL.

Estimation of Anthropogenic CH₄ and CO₂ Emissions in Taiyuan-Jinzhong Region: One of the World's Largest Emission Hotspots

Cheng Hu^{1,2} , Wei Xiao^{2,3}, Timothy J. Griffis⁴, Qitao Xiao⁵, Shumin Wang⁶, Yuzhong Zhang^{7,8}, Weifeng Wang¹, Lingyun Zhu⁶, Xin Chen⁹, Xueying Yu⁴, and Xuhui Lee¹⁰

¹College of Biology and the Environment, Joint Center for Sustainable Forestry in Southern China, Nanjing Forestry University, Nanjing, China, ²Collaborative Innovation Center on Forecast and Evaluation of Meteorological Disasters (CIC-FEMD), Nanjing University of Information Science & Technology, Nanjing, China, ³Center on Atmospheric Environment, International Joint Laboratory on Climate and Environment Change (ILCEC), Nanjing University of Information Science & Technology, Nanjing, China, ⁴Department of Soil, Water, and Climate, University of Minnesota-Twin Cities, Minneapolis, MN, USA, ⁵Key Laboratory of Watershed Geographic Sciences, Nanjing Institute of Geography and Limnology, Chinese Academy of Sciences, Nanjing, China, ⁶Shanxi Province Institute of Meteorological Sciences, Taiyuan, China, ⁷Key Laboratory of Coastal Environment and Resources of Zhejiang Province, School of Engineering, Westlake University, Hangzhou, China, ⁸Institute of Advanced Technology, Westlake Institute for Advanced Study, Hangzhou, China, ⁹Guang'an Vocational & Technical College, Guangan, China, ¹⁰School of the Environment, Yale University, New Haven, CT, USA

Abstract Coal mining ranks as the largest anthropogenic CH₄ source in China with emission factors (EFs) varying up to 30-fold among inventories when applied to different provinces. The lack of independent evaluation of coal mining CH₄ EFs in China is one of the main uncertainties in estimating national total CH₄ emissions. Shanxi province, which supplies 25% of the national coal production, is the largest coal mining CH_4 emission region in China and even among the world's largest coal production regions. This area is also a significant anthropogenic CO₂ source because of high-density power and industrial activities. Given the large uncertainties in CH₄ and CO₂ inventories from provincial to city scales, questions remain whether state-of-theart inventories have accurately estimated these emission hotspots. Here, we evaluate CH_4 and CO_2 emissions from one of the world's largest coal production regions near Taiyuan City, the capital of Shanxi province, China. CH₄ and CO₂ concentrations were measured from March 2018 to February 2019 from a 30-m tower. These data were used within an inverse modeling framework to simulate both CH4 and CO2 concentrations and to evaluate EFs for this region. Results show generally good agreement between observed and simulated CH_4 concentrations. However, the CO₂ simulations were much lower compared to the observations. Given the minor role of NEE-induced CO_2 enhancements, we believe that the large difference is attributed to the underestimation of anthropogenic CO_2 emissions. In general, the derived *posteriori* anthropogenic CH_4 emissions were $85.2(\pm 18.1)\%$ of a priori emissions, where fugitive CH₄ from coal mining accounted for ~92.7% of total anthropogenic emissions. The derived coal mining EF was $23.2(\pm 4.9)$ m³ CH₄/ton coal, close to the default value of high CH₄-content coal, but twofold the province average that were reported by previous observation-based studies in Shanxi province, indicating large spatial inhomogeneity in the coal mining CH_4 EF. The *posteriori* CO_2 emissions were 1.6-fold of the a priori emissions, highlighting underestimation of CO_2 emissions in industrial cities and some potential large emission sources that are missing from state-of-the-art inventories. Finally, we also emphasize the use of satellite observations and denser tower-based networks are essential in resolving the spatial inhomogeneity of greenhouse gas emissions.

Plain Language Summary The understanding of anthropogenic CH_4 and CO_2 emissions is basis for climate mitigation especially for global top emitting countries, but the largest issue before addressing above question is that many previous studies have found considerable bias of greenhouse gas emission for almost all inventories from city to regional scales. These facts hindered the government to make and evaluate corresponding mitigation policies. Here, to quantify CH_4 and CO_2 emissions at one of global largest CH_4 and CO_2 hotspot in China, we conducted 1 year tower-based atmospheric CH_4 and CO_2 concentration measurements and used atmospheric inversion method to constrain and evaluate their emissions, we found coal mining CH_4 emission factor has less bias but CO_2 emissions were underestimated by 1.6-fold, highlighting underestimation of CO_2 emissions in industrial cities and some potential large emission sources that are missing



from state-of-the-art inventories. Our findings indicate more work is needed for urban government to fully understand their greenhouse gases emissions.

1. Introduction

Methane (CH₄) and carbon dioxide (CO₂) are the main anthropogenic greenhouse gases contributing to global warming. Together they accounted for ~81% of direct anthropogenic radiative forcing (Seto et al., 2014). Hence, the implementation of emission reduction policies for both CH₄ and CO₂, especially in the top emitting countries, is needed to mitigate future climate change. China is the largest emitter for anthropogenic CH₄ and CO₂, with annual emissions reaching 0.065 Gt for CH₄ and 12 Gt for CO₂ in 2020. The largest anthropogenic sources for CH₄ were fugitive emissions from coal mining which accounted for 31% of national total emissions, and power industry accounted for 41% in anthropogenic CO₂ emissions (Crippa et al., 2021). In 2021, China has made a commitment to reduce CH₄ emissions under the Glasgow Agreement. Further, an ambitious commitment was also made in 2020 to reach a CO₂ emission peak by year 2030. To achieve these emission reductions, a national plan has been developed using a hierarchal approach from the national, regional, and local scales and includes different units such as provinces, cities, and specific facilities. Accurate quantification of emission changes is the basis to evaluate whether these commitments are effectively fulfilled. However, large uncertainties in estimating CH₄ and CO₂ emissions from local to regional scales may hinder these ambitious emission reduction goals (Han, Zeng, Oda, et al., 2020; Lin et al., 2021; Seto et al., 2014).

Activity data and emission factor (EF)-based "bottom-up" inventories are treated as baselines for local government to evaluate emission reduction rates. Many recent studies have reported large uncertainties in almost all state-of-the-art CH_4 and CO_2 inventories for China, where relative difference of these inventories can be >200% at city scale, especially for industrial cities, with averages of -57%–162% at provincial scales among different sectors (Han, Zeng, Oda, et al., 2020; Han, Zeng, Zhang, et al., 2020; Lin et al., 2021). Uncertainties in CH_4 emissions are comparable with CO_2 for some specific sectors such as waste treatment, with a relative difference larger than 2 times for China (Lin et al., 2021). Further, the underlying anthropogenic CH_4 emissions and its changes were still under debate with different conclusions including largely increased at the rate of 1.1 Tg/year (Miller et al., 2019), slightly increased by 0.1–0.4 Tg/year (Lu et al., 2021; Saunois et al., 2020; Sheng et al., 2021), and slightly decrease (Gao et al., 2020; Liu et al., 2021; Sheng et al., 2019).

One of the main uncertainties in estimating CH_4 emission is caused by underground coal mining. Such emission is mainly from ventilation shafts, where atmospheric air was pumped into the mines to keep lower CH_4 concentration (<0.5%) and to avoid dangerous conditions in the workplace. Hence, emissions can be affected by burial depths, different CH₄-content coal, mining methods/technologies, and the utilization rate. The available CH₄ EFs from underground coal mining for China were mainly based on nationwide measurements prior to the year of 2006 for ~800 underground mines, and are highly spatially heterogeneous (Sheng et al., 2019; Zhu et al., 2017). The province-based EFs considerably varied by 30-fold from smaller than 1 m³ CH₄/ton coal in Beijing to greater than 30 m³ CH₄/ton coal as Chongqing. Further, observations also found that EFs largely varied within the same province by sevenfold (Sheng et al., 2019). While these field measurements provided important EFs, information for specific regions remain very limited. Further, improved coal mining technologies can cause significant bias in EFs when estimating recent national CH₄ emissions (Lin et al., 2021; Zhu et al., 2017). Coal mining CH₄ EFs used for China have shown large discrepancies among different inventories, i.e., the EDGAR v4.2 was found to overestimate EFs in China by a factor of 2 for the coal mining because it applied the European averaged EFs (Saunois et al., 2016). These disagreements indicate that a robust approach is needed to monitor anthropogenic CH_4 emissions, especially from the coal mining. It can help to fill the knowledge gap in understanding their spatiotemporal evolutions, and provide independent constraint on regional CH_4 emissions.

Urban areas are also emission hotspots for both CH_4 and CO_2 and account for 21% and 70% of global anthropogenic emissions, respectively (Marcotullio et al., 2013; Seto et al., 2014). Such contributions are thought to be much larger in Northern China considering there are more anthropogenic emission sources concentrated in industrial cities and coal production areas. The above facts make these regions to be the most important target region to conducting emission mitigation policies for China. Hence, understanding the main sources and its underlying driving factors in these industrial cities is urgently needed for mitigation policy making and application. Among all provinces in China, Shanxi is a representative province which not only provide 25% (8.93 × 10⁸ tons) of



national total coal production in 2018, even larger than the world's second largest coal production country India $(7.45 \times 10^8 \text{ tons})$. Taiyuan City and Shanxi province are also among the most severely polluted regions on earth because of the high density of heavy industry, coal combustion, and anthropogenic CO₂ emissions (Mu et al., 2019; Wei et al., 2018). A recent bottom-up study found five of the top 10 largest Chinese CH₄ emission cities were located in Shanxi province (Wang et al., 2021). These combined factors indicate this region is one of the largest hotspots in China and even the world for both anthropogenic CH₄ and CO₂ emissions.

Tower-based atmospheric inversions provide a useful tool to independently evaluate local emissions and reduce corresponding inventory uncertainty (Gurney et al., 2021; Le et al., 2021; Sargent et al., 2018; Turner et al., 2020). To our best knowledge, no study has used tower-based concentration and atmospheric inverse modeling to evaluate coal mining EFs in China. Here, we constrain anthropogenic CH_4 and CO_2 emissions and provide assessment of inventory uncertainty for CH_4 emission leakage from coal mining in one of the largest anthropogenic emission regions of China. This study is based on 1-year continuous atmospheric CH_4 and CO_2 concentration observations from March 2018 to February 2019 using a 30-m tower in Taiyuan City, the capital of Shanxi province. The hourly concentration observations and three different inversion approaches were used to provide a robust constraint on anthropogenic CH_4 and CO_2 emissions, especially for coal mining CH_4 EFs.

2. Materials and Methods

2.1. Atmospheric CH₄ and CO₂ Concentration and Supporting Materials

Atmospheric CH_4 and CO_2 concentration measurements were conducted on a tower at the Xiaodian meteorological station (112°33′E, 37°44′N). The Xiaodian station is located between Jinzhong and Taiyuan City (Figure 1a). There are no obvious industrial sources within a 10-km radius of the observation site. The air sampling inlet was mounted at a 30-m height on the tower. The CH_4 and CO_2 concentrations were measured near continuously using a wavelength scan cavity ring-down spectroscopy-based analyzer (model G2301, Picarro Inc., Sunnyvale, CA). The measurement system was calibrated every 2 hr with standard gases from the Atmospheric Composition Observation and Service Center of the China Meteorological Administration that was traceable to NOAA/GML (NOAA Global Monitoring Laboratory) standards. The measurement cycle consisted of air sampling for 115 min followed by standard gas calibration for a period of 5 min. The details of the calibration system are described in Fang et al. (2014). In general, 97.0% of hourly CH_4 observations and 96.9% of the CO_2 observations were available for the whole year observation period.

Daily column-averaged dry air CH₄ mixing ratio (XCH₄) from TROPOMI measurements between May 2018 and April 2019 were also used to illustrate the spatial distribution of CH₄ concentration within Shanxi province. The TROPOMI retrievals are from the polar-orbiting Sentinel-5 Precursor satellite with spatial resolution of 7 km × 7 km (Zhang et al., 2020). Observations from the Xiaodian meteorological station were used to evaluate simulated meteorological fields from the Weather Research and Forecasting (WRF version 4.2.2) model including air temperature at 2 m (T_{2m}), relative humidity (RH), downward solar radiation (S_{\downarrow}), wind speed (WS), and wind direction (WD) at 10 m.

The observation site is located between Taiyuan and Jinzhong Cities as shown in Figure 1a. The main urban areas in both cities are closely connected and are likely to have a strong influence on the CH_4 and CO_2 observations. Here, we have named the core urban area in both cities as the Taiyuan-Jinzhong Metropolitan (TJM) area. Based on a recent bottom-up estimate of CH_4 emissions for all Chinese cities in the year 2015, Taiyuan ranked as the highest and Jinzhong as the ninth highest CH_4 emission city in China (Wang et al., 2021). Hence, the TJM area is likely the highest anthropogenic CH_4 emission urban area in China. The population of the TJM area was 8.7 million in 2020 and raw coal production was 12.18×10^7 tons in the year 2018. Shanxi province is surrounded by four provinces including Inner Mongolia, Hebei, Henan, and Shaanxi (Figure 1b). The raw coal production in these five provinces accounted for 74% of the national total production for China in 2018. It ranked as the world's greatest hotspot for CH_4 emissions from coal production and also for total anthropogenic CH_4 emissions. The coal productions within Shanxi province were obtained from Shanxi Statistical Yearbook 2018–2019 (Shanxi Provincial Bureau of Statistics, 2018–2019).

2.2. Simulation of CH₄ and CO₂ Concentration

In general, atmospheric CH_4 (and CO_2) at the observation site is the sum of background air (i.e., when an air mass enters the study domain) and the enhancement contributed by regional sources/sinks within the study domain.





Figure 1. (a) Land use map for Taiyuan-Jinzhong Metropolitan (TJM) area and the figure for tall tower, (b) domain setups in WRF-STILT model, and meteorological fields comparisons for (c) 2-m air temperature, (d) wind speed, (e) wind direction, (f) relative humidity, (i) downward shortwave radiation. Note D-STILT: STILT model domain, D1: outer nested domain in WRF model, D2: inner nested domain in WRF model, red symbol "x" and " Δ " represent tall tower location and WLG observation site, respectively, other two sites of Beijing (BJ) and Xianghe (XH) are also labeled.

The enhancements are simulated by coupling regional CH_4 (and CO_2) flux information with dynamic concentration footprint estimates that considers the atmospheric transport process within the study domain. The simplified equations for CH_4 (and CO_2) concentration simulations can be expressed as

$$CH_{4_sim} = CH_{4_bg} + \sum_{i=1}^{168} f_i \times \left(E_{CH_{4_anthro}} \right)_i + \sum_{i=1}^{168} f_i \times \left(E_{CH_{4_wetland}} \right)_i$$
(1)

$$CO_{2_sim} = CO_{2_bg} + \sum_{i=1}^{168} f_i \times \left(E_{CO_{2_bio}} \right)_i + \sum_{i=1}^{168} f_i \times \left(E_{CO_{2_anthro}} \right)_i$$
(2)

where CH_{4_sim} , CH_{4_bg} , *f*, $E_{CH_{4_anthro}}$, and $E_{CH_{4_wetland}}$ represent the simulated atmospheric CH_4 concentration, background, footprint, enhancement contributed by anthropogenic and wetland flux (Equation 1), respectively. The number of 168 hr indicates air particles are released and tracked for 7 days. The same as CH_4 for CO_2 in Equation 2.



Following our previous work (Hu et al., 2019, 2021), the CH_4 and CO_2 background values were derived from weekly sample observations at the Waliguan site (hereafter WLG, 36°17'N, 100°54'E; https://www.esrl.noaa. gov/gmd/dv/data/ last accessed: 2 January 2022). The site location is shown in Figure 1b. Here, the CCGCRV curve fitting algorithm (see Thoning et al. (1989) as recommended by NOAA, https://www.esrl.noaa.gov/gmd/ccgg/mbl/crvfit/crvfit.html) was used to calculate the hourly background values from the weekly observations

$$y = a_0 + a_1 t + a_2 t^2 + \sum_{n=1}^{4} [c_n \times \sin(2n\pi t) + \varphi_n \times \cos(2n\pi t)]$$
(3)

where y represents the CH_4 and CO_2 observations, t is time, a_0 , a_1 , a_2 , c_1 , φ_1 , c_2 , φ_2 , c_3 , φ_3 , c_4 , φ_4 are regression parameters. Here, we note that only one site has been selected to characterize the influence of background CH_4 and CO_2 concentrations for the atmospheric inversions. The associated uncertainties for this approach are discussed below.

2.2.1. Atmospheric Transport Model

The Stochastic Time-Inverted Lagrangian Transport model (hereafter STILT model) was used to simulate the hourly concentration footprint (Fasoli et al., 2018; Lin et al., 2003), which is estimated by releasing air particles at the receptor and tracking their movement backward in time and represents the sensitivity of the upstream source area to observations at the tower site (or receptor). Specifically, the footprint was calculated from the integration of released air particles and calculating their residence time in each grid box within the planetary boundary layer. Using the STILT model, we released 500 particles for each hour and tracked each particle location back in time for 7 days or until the particle left the rectangle domain (hereafter D-STILT, Figure 1b), which is centered at Shanxi province and covers an area with length around 2,000 km. The simulated hourly concentration footprint contains 7 days of atmospheric transport information, and represents the dynamic changed 168 footprints after 500 particles were released at the observation site. The WRF version 4.2.2 model was used to simulate high spatial-temporal resolution of meteorological fields to drive the STILT model. Two nested domains and a two-way feedback option were applied in WRF, with a spatial resolution of 27 km (outer nested domain, hereafter D1, Figure 1b) and 9 km (inner domain, hereafter D2), respectively. The outer nested domain covers eastern and central China, and the inner domain covers northern China, which was the largest anthropogenic CH₄ and CO₂ hotspot in China. The physical schemes used in the WRF model are the same as those used in our previous studies (Hu et al., 2019, 2021).

2.2.2. The a Priori CH₄ and CO₂ Flux

The a priori CH_4 and CO_2 fluxes are needed to simulate a priori enhancements that contain both anthropogenic and natural sources. The natural CH₄ source is mainly from wetlands, and biological CO₂ flux was emissions from biomass burning and net ecosystem exchange (NEE). The Emission Database for Global Atmospheric Research (EDGAR) provides anthropogenic CH_4 and CO_2 emissions at monthly averages with spatial resolution of 0.1°. We used the most recent v6.0 inventory for the year 2018 (Crippa et al., 2019; Janssens-Maenhout et al., 2020). Wetland CH_4 flux is from the WetCHARTs ensemble mean with 0.5° spatial resolution using monthly averages (Bloom et al., 2017). The EDGAR CH_4 inventory contains 22 different categories. The main sources are fugitive from coal mining (PRO_coal), waste water handling (WWT), and agricultural soils (AGS, mainly rice paddies). Considering that wetland CH_4 flux in WetCHARTs includes rice paddies (main source as AGS) as one wetland type, we exclude AGS in EDGAR and assume that WetCHART contains all of the anthropogenic and natural wetland CH_4 emissions. Biomass burning CO_2 flux and NEE are from the Carbon Tracker assimilation system with 3-hourly averages and 1° spatial resolution. The EDGAR anthropogenic CO₂ emissions contain 28 different categories. The main sources include power industry (ENE), combustion for manufacturing (IND), nonmetallic minerals production (NMM), and energy for buildings (RCO). Temporal profiles of the main categories were applied to CO₂ to derive hourly varied emissions (Hu et al., 2022). No temporal profiles were applied to CH_4 emissions considering that coal mining accounted for >90% of CH_4 emissions in the study region and does not have diurnal variations.

The enhancements contributed by each category and the source area are tracked as separate tagged tracers in the WRF-STILT framework. To quantify CH_4 and CO_2 enhancement contributions from different source regions, enhancement from all 11 cities in Shanxi province and surrounding four provinces as Inner Mongolia, Hebei, Henan, and Shaanxi (Figure 1b) will be simulated separately. To provide robust evaluations of bias in a priori



inventories, we also used more available inventories to compare with EDGAR v6.0 and *posteriori* emissions. They included Two different EDGAR versions (v4.0 and v4.3.2) and another independent coal mining CH₄ inventory (0.1° × 0.1°, Scarpelli et al., 2020), and four other different anthropogenic CO₂ emissions (ODIAC version 2020b (1 km × 1 km, Oda et al., 2018), PKU-CO₂ (0.1° × 0.1°, Liu et al., 2015), Carbon Tracker 2019B (1° × 1°, Peters et al., 2007), MEIC (0.25° × 0.25° Li et al., 2017)) will also be compared with EDGAR v6.0. Note only the year 2014 is available for PKU-CO₂ inventory, and year 2018 is available for other inventories.

2.2.3. Representation Error Analysis

Representation error is defined as the model setup mismatch versus the true field conditions for both horizontal and vertical resolutions (Agustí-Panareda et al., 2019; Wang et al., 2022). It also contains the observed representation error that is defined as the mismatch between point measurements and grid-cell-averaged values in models, which is usually caused by coarse spatial resolution of emissions and driving meteorological files (Gerbig et al., 2003). For example, aggregation errors are caused when aggregating heterogeneous fluxes from finer spatial resolutions into a coarser resolution. It can be expressed as the simulated enhancement difference in emissions resulting from using different spatial resolutions (Turner & Jacob, 2015; Zhao et al., 2009). The representation error can have a large influence when emissions are heterogeneously distributed, where the nearby higher (or lower) emissions in finer resolution inventory are smoothed to the regional average and can lead to a corresponding underestimation (or overestimation) of simulated enhancements (Wang et al., 2022). Theoretically, a higher spatial resolution flux map and corresponding footprint can better represent the inhomogeneity of point sources of anthropogenic emissions. Aggregation errors occur in urban areas, especially during the nocturnal period (0:00–06:00, local time) for stable boundary layer stratification. Under these conditions, atmospheric CH_4 (and CO_2) enhancements can be more easily influenced by local emissions. Here, we will analyze the aggregation errors and quantify its influence on atmospheric concentration and emission inversions in Section 3.3.

2.3. Inverse Modeling: Scale Factor Bayesian Inversion (SFBI) Method

We applied an SFBI method to interpret the tower-based CH_4 (and CO_2) observations in terms of quantitative constraints on their emissions. This Lagrangian inverse approach is an efficient way to evaluate bottom-up inventories, based on atmospheric CH_4 and CO_2 concentration observations and simulation of these concentrations by using these a priori inventories (Kim et al., 2013; Miller et al., 2008). The relationship between observed and simulated tower CH_4 and CO_2 concentrations can be expressed as follows:

$$v = K\Gamma + \varepsilon \tag{4}$$

where y is vector of observed enhancement (concentration minus background) for target gases, K corresponds to simulated enhancements from different categories or tagged area or both, Γ consists the a *posteriori* scaling factors (hereafter SFs) for each source components in K, ϵ is the observing system error.

J

The optimal way to get a *posteriori* SFs is to minimize the cost function $J(\Gamma)$, which represents the error weighted mismatch between observed and simulated CH₄ (and CO₂) enhancements and mismatch between the a priori and *posteriori* SFs, the cost function $J(\Gamma)$ is expressed as

$$J(\Gamma) = \frac{1}{2} \left[(y - K\Gamma)^T S_e^{-1} (y - K\Gamma) + (\Gamma - \Gamma_a)^T S_a^{-1} (\Gamma - \Gamma_a) \right]$$
(5)

where S_e and S_a are the observational and a priori error covariance matrices, and S_e consists of measurement and model errors. Each element of Γ_a is equal to 1. Therefore, the solution for minimizing this cost function and obtaining the *posteriori* SFs is to solve $\nabla_{\Gamma} J(\Gamma) = 0$, more details can be referred in Kim et al. (2013) and Hu et al. (2019, 2022). The "averaging kernel" will be used to quantify the sensitivity of the retrieved emissions to their true values (Kim et al., 2013), with value of 1 representing the highest contribution and sensitivity, and value of 0 of the lowest.

2.4. Constructions of Error Covariance in Bayesian Inversion Approach

We applied the same approach as previous studies (Chen et al., 2018; Griffis et al., 2017; Hu et al., 2013, 2019, 2022; Kim et al., 2013), which did not consider the off-diagonal elements of the error variance-covariance matrix. The



error covariance and state vector metrics in SFBI approach for observations (S_e) and a priori values (S_a) are specified as follows. The measurement errors lie in the accuracy of calibration strategy and background, where enhancements (y) are evaluated instead of concentration observations. Because our measurement system has high precision and was calibrated every 2 hr, here measurement errors mainly came from CH₄ (and CO₂) background. Based on comparisons of enhancements and background concentrations for CO₂ and CH₄ with different background sites (Section S5 and Table S1 in Supporting Information S1), a 10% relative uncertainty is assigned to measurement error. The model error contains two primary sources as representatives of released particles numbers ($S_{particles}$) in STILT model, and uncertainties in meteorological field simulations (S_{met}) especially from wind speed, wind direction, and boundary layer height (hereafter PBLH). Following previous work, an uncertainty value of 13% was assigned to $S_{particles}$ (Chen et al., 2018; Hu et al., 2022; Kim et al., 2013), and 20% was assigned to S_{met} based on comparisons with midday PBLH with Carbon Tracker simulations and evaluations of wind speed and directions (Gourdji et al., 2018; Kim et al., 2013).

Note our WRF-STLIT model setup can simulate enhancement contributions from different categories and regions as displayed below, but considering the limited ability of SFBI approach in resolving SFs for both categories and regions, we only derive *posteriori* SFs for the main emission categories following previous studies (Griffis et al., 2017; Hu et al., 2019, 2022). For the uncertainty in the a priori flux, previous studies have compared different inventories and found the anthropogenic CH₄ differences were found to vary between 50% and 200% (Han, Zeng, Oda, et al., 2020; Han, Zeng, Zhang, et al., 2020; Lin et al., 2021), and CO₂ emission differences in Chinese industrial cities can reach to 200%. Because coal mining is the largest contributor to CH₄ in TJM area, we applied a 3-element (coal mining, reset anthropogenic sources, and wetland) state vector in for constraining CH₄ emissions. Considering the potential uncertainty of city scale coal mining CH₄ emissions can reach to 200%, and also to analyze whether the derived *posteriori* SFs for coal mining CH₄ emissions were sensitive to a priori uncertainty, we applied three uncertainties combination cases, where the uncertainty were 50% (Case 1), 100% (Case 2), 200% (Case 3) for coal mining, respectively. And all three cases assigned 200% for the rest anthropogenic categories and wetland which together only accounted for <10% of regional CH₄ total emissions.

To constrain anthropogenic CO_2 emissions, we first applied 4-element (ENE, NMM, others, biological sources) state vector and 3-element (ENE, others, biological sources) state vector. But our preliminary test results in TJM area indicated there should be missing of large CO_2 hotspots that have not been considered in EDGAR v6.0 inventory. Besides, results from above setups also indicated the spatial distributions of different anthropogenic CO_2 categories are not accurate. All these factors lead to the condition that there is not enough information from the observations to resolve many elements and the exact locations of missing CO_2 sources, and anthropogenic CO_2 categories cannot be resolved by single factor Bayesian inversion approach in this study. Here, we finally choose to use a 2-element (anthropogenic sources, biological sources) state vector following the study by Hu et al. (2013) and Kim et al. (2013), which does not require the partition of all anthropogenic emissions into different sources. We found that the SFs were not sensitive to a priori uncertainty and finally choose 200% for anthropogenic CO_2 emissions and 100% for biological flux based on the concentration comparisons in Section 3.3.

3. Results and Discussions

3.1. Observed CH₄ and CO₂ Concentration Variations

We first evaluated the WRF model simulated meteorological fields (including T_{2m} , RH, S_1 , WS, and WD at 10 m) with observations at the Xiaodian station (Figure 1c). The comparisons showed relatively good WRF model performance (see Table S2 in Supporting Information S1) and indicate that the WRF simulated meteorological fields can be used to drive the STILT atmospheric transport for CH₄ and CO₂. Figure 2a shows hourly concentration observations of both CH₄ and CO₂ and reveals relatively strong seasonal patterns with CH₄ and CO₂ reaching maximum values from November to January. This pattern is likely driven by a combination of higher emissions, lower PBLH variations, and more stable boundary layer stratification during this period. Here, the local home heating period general occurred from 1 November 2018 to 31 March 2019, indicating higher CO₂ emissions from coal and natural gas combustion. The seasonal concentration averages (enhancements are shown in parentheses) for CH₄ were 2466.8 ppb (555.0 ppb), 2562.2 ppb (624.1 ppb), 2616.7 ppb (692.9 ppb), 2701.6 ppb (777.8 ppb) in spring, summer, autumn, and winter, respectively; and were 436.9 ppm (22.6 ppm), 431.4 ppm (26.1 ppm), 453.5 ppm (44.6 ppm), 465.1 ppm (51.8 ppm) for CO₂ during the same periods. The results demonstrate that CO₂ enhancements in autumn and winter were >2 times that in spring. Further, these data also indicate that the



21698996

2023, 8, Downlo



Figure 2. (a) Time series of hourly CO_2 and CH_4 observations, and (b) scatter plots of observed CH_4 and CO_2 enhancement, each point is midday averages (12:00–18:00, local time), color bar indicates the corresponding months for each point.

seasonality in CO_2 flux was larger than for CH_4 when considering that both gases were subject to similar atmospheric transport processes and PBLH variations.

Note the enhancement ratio between two different gases is considered as their emission ratios (Tohjima et al., 2020; Turnbull et al., 2011, 2015; Vardag et al., 2015), we further displayed the relationship between observed midday CH₄ and CO₂ enhancement in Figure 2b, they show strong linear relationship (R = 0.78, P < 0.001) with slope of 12.2 (±0.7) (ppb/ppm). The linear slope indicates the emission ratio between CH₄ and CO₂ was around 0.0122, this slope will be compared with model simulations to examine bias in the a priori emissions in Section 3.6.

To evaluate whether the CH_4 and CO_2 concentration was comparable with other cities, we further compared the annual averages of CH_4 and diurnal averages of CO_2 with some typical urban area in China (Beijing, Nanjing, Xianghe) and other countries including three cities (London, Baltimore/Washington, Los Angeles) for CH_4 and four cities for CO_2 (Salt Lake City, Los Angeles, Boston, and Indianapolis, Figure S2 and Section S2 in Supporting Information S1). The comparisons illustrate midday CO_2 concentration at our observation site is one of the

8 of 22



highest and only slightly lower than Beijing. Given the lack of observed diurnal CH_4 variations reported in previous studies, we compared our results to annual averages. In these previous studies, all CH_4 observations were calibrated frequently with standard gases that can be traceable to NOAA/GML (NOAA Global Monitoring Laboratory) standards. We believe their absolute accuracy is better than 1 ppby, which can be the largest systematic bias. Considering the magnitude of CH_4 enhancements at our site is about 2–3 times larger than other major cities, the systematic bias could not explain the large enhancement difference between our study and above-mentioned cities. These comparisons indicate strong CO_2 and CH_4 emissions from the TJM area compared to other major cities.

3.2. The a Priori Flux and Simulated Source Footprint

3.2.1. The a Priori CH₄ and CO₂ Flux

The a priori anthropogenic CH₄ emissions from EDGAR v6.0 and WetCHART wetland emissions are displayed in Figures 3a and 3c. The main anthropogenic sources are located in the central and southern Shanxi province. Wetland emissions were relatively low when compared with the anthropogenic emissions. We also show the satellite-based CH₄ column concentration data in Figure 3b. These data show strong coherency (R = 0.36, P < 0.001, Figure S6 in Supporting Information S1) with the EDGAR v6.0 CH₄ inventory in Shanxi. The spatial distribution of the high CH₄ column concentration data (>1,900 ppb) is also consistent with the location of coal mines (Figure S3 in Supporting Information S1), and indicates that coal mining CH₄ dominated the regional atmospheric CH₄ distributions. The a priori anthropogenic CO₂ emissions from EDGAR v6.0 and ODIAC (other inventories not shown), and NEE are displayed in Figures 3d–3f, respectively. The annual averaged NEE was negligible when compared with the anthropogenic emissions. However, we should also note the hourly NEE may also be comparable with anthropogenic emissions for some specific days in summer which will be displayed below. Further, CO₂ emissions from within the TJM area represent a significant hotspot (i.e., values are in the upper 10% percentile).

Considering the EDGAR inventories have been widely used as a priori emission in many previous studies, with the highest spatial resolution in China for both CH_4 and CO_2 emissions, and the spatial distributions of satellite-based CH_4 concentrations also showed high consistence with EDGAR emissions in Shanxi province, indicating the general good spatial distributions of anthropogenic CH_4 especially for coal mining. Hence, only EDGAR v6.0 inventory was used in inversion for both CO_2 and CH_4 emissions in this study.

The EDGAR v6.0 anthropogenic CH_4 and CO_2 emissions from the top five emission sources for both TJM area and Shanxi province are displayed in Table 1. We found that CH_4 emissions from coal mining accounted for >90% of the total anthropogenic CH_4 emissions in both the TJM area and Shanxi province. The second most important source was waste water, which accounted for <4% of emissions. These a priori emissions proportions indicate that the CH_4 enhancements at our observation site are dominated by coal mining CH_4 emissions. We also calculated the coal mining CH_4 emissions in Shanxi province using a different inventory available for the year 2016 by Scarpelli et al. (2020), where the emission was 7.4×10^6 tons and 88% of 8.4×10^6 tons for EDGAR v6.0 in the same year. Previous studies have found large discrepancies in the coal mining CH_4 emissions and its mean EF for Shanxi province, with observed mean EF values ranging from 5.6 to 12.7 m³ CH₄/ton coal and largely varied from 3.3 to 22.1 m³ CH₄/ton coal for different coal quality types within Shanxi (Table S3 in Supporting Information S1). We derived the EFs by using three different EDGAR versions (v4.0, v4.3.2, v6.0) in the year 2010 (the most recently released versions for all three EDGAR versions). The EFs decreased from 16 m³ CH₄/ton coal for v4.0 to 12 m³ CH₄/ ton coal for v6.0 in the same year 2010, note most of EFs applied by EDGAR inventories are IPCC recommended default values, but EGDRA will update them for the regions with localized EFs (Solazzo et al., 2021). Hence, we assume that such discrepancies will lead to large uncertainty in the estimated CH₄ emissions by bottom-up methods.

The EDGAR v6.0 CO₂ emissions from the power industry accounted for ~60% of the total anthropogenic CO₂ emissions in TJM area and 50% for Shanxi province, indicating large emissions from coal burning-based power stations. The second and third largest sources were cement production and combustion for manufacturing. We also calculated the annual anthropogenic CO₂ emissions from EDGAR v6.0, ODIAC, MEIC, and PUK-CO₂ inventories, which were 5.1×10^8 tons, 3.2×10^8 tons, 4.9×10^8 tons, and 3.9×10^8 tons for Shanxi province, respectively. The anthropogenic CO₂ emissions in TJM area were 9.3×10^7 tons, 6.2×10^7 tons, and 7.8×10^7 tons from EDGAR v6.0, ODIAC, and PKU-CO₂. Here, the MEIC and Carbon Tracker emissions are not shown because of its coarse spatial resolution at the city scale. As evaluated by Han, Zeng, Oda, et al. (2020) and Han, Zeng, Zhang, et al. (2020) for the anthropogenic CO₂ emissions in Shanxi province based on seven different inventories,



Journal of Geophysical Research: Atmospheres



Figure 3. (a) EDGAR v6.0 CH₄ emissions, units: \log_{10} (mol m⁻² s⁻¹), (b) column-averaged CH₄ concentration from Sentinel satellite retrievals, units: ppb, (c) wetland CH₄ emissions, units: \log_{10} (mol m⁻² s⁻¹), (d) EDGAR v6.0 CO₂ emissions, units: \log_{10} (mol m⁻² s⁻¹), (e) ODIAC CO₂ emissions, units: \log_{10} (mol m⁻² s⁻¹), (e) annual averaged net ecosystem exchange (NEE), units: mol m⁻²s⁻¹, (g) annual average of hourly footprint for CO₂/CH₄ observation site, (h) enlarged figure of footprint.

emissions varied from $\sim 3.0 \times 10^8$ tons to $\sim 7.5 \times 10^8$ tons in year 2012 by 2.5 times, which also indicated relatively large uncertainty in Shanxi province and even larger uncertainty at the city scale.

3.2.2. Tower-Based Concentration Footprint and Source Regions

The WRF-STILT simulated footprint represents potential source area variations for each hourly observation. The climatological source footprint was obtained by averaging the hourly values over the whole year (Figures 3g

The a Priori Anthropogenic CO₂ and CH₄ Emissions for TJM Area and Shanxi Province in Year 2018 From EDGAR v6.0 Inventory

			CH_4 (×10 ³ tons)						$CO_2 (\times 10^6 \text{ tons})$						
		PRO_COAL	WWT	SWD_LDF	RCO	ENF	Others	Total	ENE	NMM	IND	TRO	RCO	Others	Total
Taiyuan	Emission	642.1	27.8	19.6	9.1	3.0	8.1	709.7	31.0	5.2	5.2	2.9	2.3	4.6	51.2
	Proportion	90.4%	3.9%	2.8%	1.3%	0.4%	1.1%	100.0%	60.6%	10.1%	10.1%	5.6%	4.6%	9.0%	100.0%
Jinzhong	Emission	1584.8	17.6	9.2	6.5	4.2	35.0	1657.5	25.3	5.7	2.5	0.5	2.1	5.3	41.3
	Proportion	95.5%	1.1%	0.6%	0.4%	0.3%	2.3%	100.0%	61.0%	13.8%	6.0%	1.3%	5.1%	12.8%	100%
Shanxi	Emission	8325.7	260.3	142.0	92.6	107.8	134.4	9062.7	267.3	65	67.7	17.9	20.6	720.4	510.7
	Proportion	91.9%	2.9%	1.6%	1.0%	1.2%	1.5%	100.0%	52.3%	12.7%	13.3%	3.5%	4.0%	14.1%	100.0%

and 3h, enlarged on the right). The most intense source footprint area (i.e., values greater than 10^{-3} ppm m² s/ µmol (displayed in yellow)) includes the TJM area. Further, the tower-based observation site is located in the middle of TJM area, indicating that our CH₄ and CO₂ observations are representative of the anthropogenic signals originating from within the study domain (Figures 1a and 1b).

To quantify the source area representation of observed CH_4 and CO_2 concentration, we calculated how many hours were needed to accumulate different proportions of total enhancements, where simulated CO_2 enhancements were used because they reflect the hourly variation of emissions. The temporal dynamics of the cumulative anthropogenic CO_2 enhancements are shown in Figure S4 in Supporting Information S1 for all four seasons and



Figure 4. Monthly mean enhancement contributions from Taiyuan, Jinzhong, Shanxi province, and other regions for (a) CH_4 and (b) CO_2 .

are discussed in details in Section S3 in Supporting Information S1. There are key differences in the time of day and time of year required to accumulate the same enhancement proportion. The average time to accumulate a 50% of total enhancement was 3 hr in near midnight and 6–12 hr near midday for all seasons. We further used annual wind speed average and the accumulated time to calculate the source area by atmospheric transport, where observed wind speed of 2.6 m/s in the midnight and 3.5 m/s in the midday were used. Results indicate that the 50% enhancement source area was with radius of 28 km in midnight and 76–152 km in midday. The midday radius was comparable with the size of TJM area, where the length and width varied by 100–200 km, but the 28-km radius for midnight only represents local emission sources.

To further quantify the enhancement contributions from different source regions (i.e., TJM area, Shanxi province, and other provinces), enhancements were tracked as separate tagged tracers within the WRF-STILT framework. The daily variations are displayed in Figure S5 in Supporting Information S1 and the monthly mean values are shown in Figures 4a and 4b. We found that the CH₄ enhancement (CO₂ shown in parentheses) proportions had a strong seasonal pattern, which increased from the lowest value 74.6% (59.6%) in summer to a maximum of 93.1% (82.3%) in winter for Shanxi province's contributions and from 38.2% (41.0%) to 62.6% (64.3%) for the TJM area. The annual mean CH₄ enhancement proportions were 85.0%, 1.3%, 1.7%, 2.6%, and 1.9% for Shanxi, Inner Mongolia, Hebei, Henan, and Shanxi province, respectively. The CO₂ enhancement proportions were 71.6%, 5.2%, 4.5%, 4.2%, and 3.7% for corresponding provinces. The TJM area accounted for 51.8% (54.5%) of the CH₄ and CO₂ enhancements. Here, Taiyuan accounted for 36.8% (35.3%) and Jinzhong accounted for 15.0% (19.2%).

3.3. Representation Error Analysis

Before constraining the anthropogenic CH_4 and CO_2 emissions, we assessed the representation errors in the model simulation and their potential influence on derived *posteriori* SFs. Instead of using all-day observations in the



atmospheric inversion study, previous studies have found that atmospheric transport simulations are likely more reliable for greenhouse gas concentration during midday compared to early morning or nocturnal periods. For example, the evaluation of five different Eulerian atmospheric transport models with multiple observation sites indicated that the model performed better in the daytime because of well-mixed atmospheric conditions than under stable nighttime conditions (Geels et al., 2007). In Section S4 and Figure S7 in Supporting Information S1, we quantified and discussed the representation error caused by both spatial resolution (aggregation error) and vertical resolution between midday and midnight. We first calculated the relative difference of simulated midday anthropogenic CO₂ enhancement between using 0.1° and 1° EDGAR v6.0 CO₂ emissions, which were 7.9%, 4.4%, 19.1%, and 20.2% for spring, summer, autumn, and winter, respectively. Similarly, the differences were 0.8%, 26.9%, 11.5%, and 19.0% for simulated anthropogenic CH_4 enhancements. The relative difference of simulated midnight anthropogenic CO₂ enhancements between using 0.1° and 1° EDGAR v6.0 emissions were 1.5-2 times and were up to 6 times for CH_4 enhancement simulations. The midday relative difference was much smaller than using midnight data and indicate that the midday data can substantially minimize the aggregation error. Note although much finer spatial resolution of emissions (i.e., 1-km spatial resolution of ODIAC inventory) can better represent spatial distributions of CO2 emissions, it still needs the same spatial resolution of WRF-STILT model simulated footprints, which is >100 times of computation resources than using 0.1°. Here, we only compared the aggregation errors by using 0.1° and 1° to keep the balance between deriving conclusions and computation cost.

The relatively large differences at midnight are likely caused by: (a) there were strong point sources within a 1° radius (~100 km) of the observation site (Figure 1). Aggregation from 0.1° to 1° emissions will act to smooth higher emissions that is much further/closer with observation site to regional averages; (b) lower PBLH and stable stratification at midnight will assign larger footprint weight to emissions sources that are closer to the tower; (c) the STILT model assumes that all emissions are surface sources. However, there can be strong point sources (i.e., stack height and subsequent plume rise height for coal burning power station CO_2 and coal mining CH_4 emissions), which will lead to overestimation of simulated concentrations when the PBLH is below the emission height. But the emission height is less important when atmospheric mixing is strong during midday and the PBLH is considerably higher than the stack height (Brunner et al., 2019). A recent study comparing emissions with or without using emission height in WRF-STILT found that their differences were negligible near midday (Maier et al., 2022). Based on these analyses, we conclude that CH_4 and CO_2 observations in the midday have smaller representation errors and will be used here for comparison with observed tower concentrations to constrain the emissions.

3.4. Comparisons Between Observed and Simulated Midday CH4 and CO2 Concentrations

Time series of observed and simulated midday CH_4 and CO_2 concentrations are shown in Figures 5a and 5b. In general, the annual midday averages for observations and simulations were 2340.9 and 2301.5 ppb for CH_4 , 436.7 and 422.7 ppm for CO_2 . The statistical analyses of *R* (correlation coefficient), RMSE (root mean square error), and MB (mean bias) were 0.41, 384.7, and 39.5 ppb for CH_4 and were 0.69, 22.4, and 14.0 ppm for CO_2 . The lower *R* for CH_4 than CO_2 and the more scattered enhancements found in Figures 5c and 5d indicate the spatial heterogeneity of CH_4 emissions is not as well resolved for CO_2 in the EDGAR (0.1° spatial resolution) inventory. More detailed and updated information are needed to further improve its spatial resolution. Previous studies also found large bias in the numbers and locations of coal mining for present inventories in China (Lin et al., 2021; Zhu et al., 2017). The simulated CO_2 concentrations (and enhancements) were much smaller than the observations as displayed in Figure 5b. CH_4 did not show such overestimate/underestimate. Regression slopes of observed and simulated enhancements were $1.12 \pm (0.11)$ for CH_4 and $0.55 \pm (0.04)$ for CO_2 (Figures 5c and 5d), also supporting that the simulated CO_2 enhancement was underestimated. Such large discrepancy for CO_2 could be caused by the following factors: (a) The PBLH was overestimated in the WRF-STILT model and caused the underestimation of CO_2 uptake by plants; and (c) a combination of both factors.

To help address these concerns, CO_2 simulations from the Carbon Tracker model was added to Figure 5b and PBLH simulations using two models were also compared in Figure S8 in Supporting Information S1. Here, we used Carbon Tracker because it is widely used to simulate global CO_2 distributions. Further, it is an Eulerian atmospheric transport model, which takes a different approach than the STILT model. Hence, the comparison of two different models with observations can provide us with a robust evaluation of potential bias in the a priori emissions and performance of model simulations. These comparisons indicate that: (a) Both WRF-STILT and

21698996





Figure 5. Time series of observed and simulated midday (a) CH_4 concentrations, (b) and CO_2 concentrations, note CO_2 simulations from Carbon Tracker model is also displayed, and each dot represent daily averages; (c) scatter plots between observed and simulated CH_4 and (d) CO_2 using data excluded statistical outliers (>98th percentile in simulations), uncertainty values indicate 95% confidence limits; and simulated midday enhancement components for (e) CH_4 and (f) CO_2 with anthropogenic and biological enhancements displayed.



Carbon Tracker simulations show similar daily variations with CO_2 observations, indicating these models can well simulate weather patterns and the atmospheric transport process. (b) The intermodel comparisons only show slightly higher CO_2 simulation in WRF-STILT than Carbon Tracker. This is likely because each Carbon Tracker grid cell represents averages of $2^{\circ} \times 3^{\circ}$ area, such that the aggregation error will smooth out the higher emissions within the Taiyuan-Jinzhong urban area (with area of $<0.5^{\circ} \times 0.5^{\circ}$ in Figure 1a). The grid-cell resolution in WRF-STILT is $0.1^{\circ} \times 0.1^{\circ}$ and can better resolve the spatial distribution of anthropogenic emissions. Further, because only midday averages are used, the aggregation error is minimized and cannot lead to such large underestimation. (c) The PBLH in both models showed high consistency, although the WRF simulated PBLH was lower than Carbon Tracker for winter. However, this would be expected to result in higher CO_2 concentrations and cannot explain the underestimation of CO_2 value. (d) The comparisons between observed and simulated CO_2 enhancement illustrated underestimation of simulated midday CO_2 enhancement occurred in most study period of four reasons for both models, even though the source contribution from TJM area changed from ~10% to ~100% (Figure 4 and Figure S7 in Supporting Information S1, discussed above), these analysis indicate such bias exist when air flows from different directions and represent common bias in a priori emissions at regional scale.

We also displayed the enhancement components from each category for both CH_4 and CO_2 (Figures 5e and 5f). These results show that anthropogenic sources dominated the seasonal variations compared to wetland source of CH_4 and NEE for CO_2 . Simulated annual a priori CH_4 enhancement averages were 374.6 ppb for anthropogenic sources and 1.6 ppb for wetlands. Simulated annual a priori CO_2 enhancement averages were 12.7 ppm for anthropogenic sources and -0.7 ppm for NEE. Based on the above analysis, we concluded that the underestimation of simulated CO_2 concentration throughout the year cannot be explained by PBLH biases, but is more likely caused by the underestimation of anthropogenic CO_2 emissions in the a priori inventory. In the next section, we will use the SFBI method and atmospheric CH_4 and CO_2 observations to constrain their emissions and derive the *posteriori* emissions.

3.5. Constraint of Anthropogenic CH₄ and CO₂ Emissions

The time series of hourly wind direction Figure 1e indicate large hour-to-hour variability. Because our site is located near the center of the TJM area, the observations represent contributions from different sources with varied wind directions. Further, as shown in Figure 5a, the underestimation of CO_2 exists for the majority of the study period, indicating the underestimation of CO_2 emissions for the region. The simulated a priori CH_4 concentrations also showed relatively good agreement from March to October, and an overestimation from November to February (Figure 5b). Based on the above analyses, we have decided to constrain CH_4 and CO_2 emissions at the seasonal scale.

In the SFBI method, hourly CH_4 and CO_2 concentration observations at midday were used to constrain anthropogenic sources in each season. As discussed in Section 2, three cases of uncertainty combinations will be used to provide robust evaluation of emissions inversion. Here, the derived seasonal *posteriori* SFs of three cases for coal mining, other anthropogenic categories, and wetlands are displayed in Table S4 in Supporting Information S1. The high averaging kernel (>0.8) for coal mining and low averaging kernel (<0.1) for wetland indicates atmospheric CH_4 observations were more sensitive to coal mining emissions than wetland. To provide a robust constraint on the SFs, we averaged the SFs from all three cases to estimate the final *posteriori* SFs. The uncertainties were represented by the range of SF estimates. The *posteriori* SFs were $0.94(\pm 0.06)$, $1.08 (\pm 0.08)$, $0.80(\pm 0.23)$, and $0.80(\pm 0.31)$ for coal mining in spring, summer, autumn, and winter, respectively; and were $1.86 (\pm 0.07)$, $1.44 (\pm 0.09)$, $1.02 (\pm 0.15)$, and $0.96(\pm 0.08)$ for the other categories.

The *posteriori* results only show slight seasonality for anthropogenic CH_4 emissions, because this area is dominated by fugitive CH_4 from coal mining. Further, home heating in winter only played as small role in the total regional CH_4 emissions. After resimulating CH_4 concentrations by applying these *posteriori* CH_4 emissions, the regression slopes for CH_4 concentrations decreased from 1.12 (±0.11) to 0.92 (±0.08), and the *y*-intercept changed from -354.3 to 100.4 ppb (Figure 6a). The time series comparisons between a priori and *posteriori* CH_4 concentrations also showed that the overestimation of a priori emissions from November to February was significantly improved following the inverse optimization (Figure S9a in Supporting Information S1). Further, the RMSE decreased from 384.5 to 339.8 ppb. After applying these SFs, the derived annual *posteriori* anthropogenic CH_4 emissions for the TJM area was $21.9(\pm 4.4) \times 10^5$ tons, which was 7.6% lower than the a priori emissions of 23.7×10^5 tons. The *posteriori* CH_4 emission from coal mining was $20.2(\pm 4.3) \times 10^5$ tons, 9.4% lower 21698996





Figure 6. Linear regression between observed and simulated (a) CH_4 and (b) CO_2 concentrations, simulations by using the a priori and a *posteriori* emission were both compared with observations, the uncertainty values indicate 95% confidence limits.

than the a priori emission of 22.3×10^5 tons. We then divided emission by the coal production of 1.22×10^8 tons in 2018 for this region, and the *posteriori* coal mining EF was $23.2(\pm 4.9)$ m³ CH₄/ton coal. Here, the coal production was obtained from local government, which conducted provincial policies as "consolidation to large coal mines" and "phase out of small coal mines" to increase coal production efficiency (Zhang et al., 2022), but some small coal mines have been frequently reported still under operation by local news. It indicates the real coal production may be slightly higher than public data and EF can be slightly smaller than $23.2(\pm 4.9)$ m³ CH₄/ ton coal.

The derived seasonal *posteriori* SFs for anthropogenic CO₂ emissions were 1.70 ± 0.02 , 1.60 ± 0.01 , 1.59 ± 0.05 , and 1.56 \pm 0.09 in spring, summer, autumn, and winter, respectively. And SFs for NEE were 0.89 \pm 0.07, 0.43 ± 0.02 , -0.40 ± 0.31 , and 1.96 ± 0.87 in corresponding seasons. The *posteriori* SF of NEE in fall indicated the sign of a priori Carbon Tracker NEE in our study region is wrong, and also with large bias in other seasons. The SFs for anthropogenic emissions indicate slight seasonal variations, where emissions in winter were the lowest. We hypothesize that the reduced industrial emissions from factories during the traditional Chinese spring festival resulted in decreased fossil fuel consumption. To evaluate whether these SFs improved the CO₂ emission estimates, we resimulated CO₂ concentration by using the *posteriori* anthropogenic/biological emissions and compared them with the CO₂ observations. These results indicate the posteriori CO₂ concentration simulations largely improved when compared with using the a priori emissions (Figure 6b). The RMSE decreased from 22.4 to 10.6 ppm, MB decreased from 14.0 to 5.6 ppm. The SFs calibrated the underestimation of simulated CO₂ concentrations. Here, the slope improved from 0.59 (\pm 0.04) to 0.82((\pm 0.04), the y-intercept decreased, and the scatter plots show less scatter (Figure 6b). The time series comparisons between a priori and *posteriori* CO₂ concentrations also showed that the underestimation of a priori emissions was calibrated reasonably well across the whole year following the inverse optimization (Figure S9b in Supporting Information S1). Based on these derived SFs for anthropogenic emissions, the *posteriori* annual anthropogenic CO₂ emissions for TJM area were $14.9(\pm 0.4) \times 10^7$ tons, which was 60.2% higher than the a priori EDGAR v6.0 inventory. Considering the a priori anthropogenic CO₂ emissions in TJM area were 9.3×10^7 tons, 6.2×10^7 tons, and 7.8×10^7 tons from EDGAR v6.0, ODIAC, and PKU-CO₂ inventories, our results show the underestimation was common among all available inventories, and indicate some large emission sources and hotspots are missing in recent inventories for this industrial region. Furthermore, to quantify the potential uncertainty of background concentration on emission inversions, we also compared the CO_2 and CH_4 concentration at WLG background site with other three backgrounds in different directions. The results indicated the uncertainty was within 5% at the annual averages for both CO_2 and CH_4 emissions, as discussed in details in Section S5, Table S1, and Figures S10–S12 in Supporting Information **S1**.





Figure 7. Diurnal averages of observed and simulated concentrations in (a) spring, (b) summer, (c) autumn, (d) winter for CO_2 , and (e) spring, (f) summer, (g) autumn, (h) winter for CH_4 .

3.6. Further Evaluation of the a Priori Emissions

The comparisons of seasonally averaged diurnal averages between observations and simulations are displayed in Figure 7. The model simulations in midday illustrated similar diurnal patterns but with different amplitude for CO₂. And simulations for CH₄ showed overall consistency with the observations. The derived ratios between observed and simulated midday CO₂ enhancements were $1.88(\pm 0.40)$, $1.81(\pm 0.46)$, $2.26(\pm 0.34)$, and $2.01(\pm 0.31)$ in spring, summer, autumn, and winter, respectively, also indicating large underestimation of anthropogenic CO₂ emissions as found by SFBI method. Further, the scatter plots between CO₂ and CH₄ concentration (and enhancement) for both observations and simulations are shown in Figure 8. Each observation represents different source areas and the linear relationship illustrates the general emission ratio between CO₂ and CH₄. We found that the regression slopes were $0.085(\pm 0.05)$ for observed CO₂ and CH₄ concentration and $0.082(\pm 0.05)$ for their enhancement ratio. For the simulations, the slopes were $0.036(\pm 0.02)$ for concentration and $0.032(\pm 0.02)$ for the enhancement ratio. Both are close to the emission slope of $0.028 (\pm 0.02)$ derived from the EDGAR a priori inventory (Figure 8b). These large discrepancies between observed and simulated CO₂ and CH₄ slopes also indicate large bias for the a priori anthropogenic CO₂ emissions.

3.7. Comparisons With Previous Studies

The comparison between the a priori EDGAR v6.0 and *posteriori* emissions is displayed in Figure 9 for TJM area. It illustrates that the *posteriori* emissions for coal mining and total anthropogenic CH₄ emissions were 92.4% and 90.6% of a priori EDGAR v6.0 inventory. Overall, there are only a few available atmospheric inversion studies for constraining coal mining CH₄ emissions in China. Miller et al. (2019) and Chen et al. (2022) used satellite observations for China, which included Shanxi province. Lu et al. (2021) and Zhang et al. (2022) used both satellite and tower-based observations to constrain CH₄ emissions. The spatial resolutions for these studies were $2.0^{\circ} \times 2.5^{\circ}$ latitude-longitude, $0.25^{\circ} \times 0.3125^{\circ}$, $4^{\circ} \times 5^{\circ}$, and $0.5^{\circ} \times 0.625^{\circ}$, respectively. Zhang et al. (2022) used seven atmospheric CH₄ sites in China (site locations were displayed in Zhang et al. (2022)), and even the closest atmospheric CH₄ tower site (SDZ site) was ~500 km (5°) from the TJM area. The concentration footprint of a tall tower based on our WRF-STILT model simulations indicates that the 50% enhancement source area was within about 100 km. Hence, we believe that the observation site used in our study is the most sensitive to both CH₄ and CO₂ emissions within the TJM area.

Among these previous studies, only Zhang et al. (2022) provided the a priori and *posteriori* CH_4 emissions from different source categories. We examined the ratio between the *posteriori* and a priori coal + gas + oil



10.1029/2022JD037915



Figure 8. (a) Comparisons of observed and simulated $CH_4:CO_2$ enhancements, (b) emission ratio between CO_2 and CH_4 with different radius of source area in EDGAR v6.0 inventory, (c) comparisons of observed and simulated $CO_2:CH_4$ concentrations, and (d) $CO_2:CH_4$ enhancement.

 CH_4 emissions for Shanxi province and TJM area. We believe their results for Shanxi province were mainly constrained by the satellite observations (not the tower observations i.e., when considering the footprint of the tower and sensitivity of a priori emissions). We found that the derived ratio was ~1 with slight spatial variations (Figure S13 in Supporting Information S1). This supports relatively good agreement at the regional scale for CH_4 emissions. Overall, our study in the central Shanxi is the first one that combined tower-based CH_4 concentration observations to constrain CH_4 emissions from coal mining in China with a much finer spatial resolution of 10 km and our results showed high coal mining EF at TJM area and no significant bias in anthropogenic total and coal mining CH_4 emissions.

To evaluate whether the derived coal mining EFs for these areas was comparable with other regions in China, we have derived the a priori EF for Shanxi and all of China and compared them with previous studies (Table 2 and Table S1 in Supporting Information S1). The *posteriori* EF for the TJM area was $23.2(\pm 4.9)$ m³ CH₄/ton coal, and was close to the high CH₄-content coal EF (22.1 m³ CH₄/ton coal), but 7 times greater than the low CH₄-content coal EF (3.3 m³ CH₄/ton coal) for Shanxi province as reported by Sheng et al. (2019). This *posteriori* EF was ~2 times the Shanxi average and 3 times the national average when compared with previous "bottom-up" observations (Lin et al., 2021; Liu et al., 2021; Sheng et al., 2019), indicating large spatial inhomogeneity in coal mining EFs in China's largest coal production province.



21698996, 2023, 8, Downle



Figure 9. Anthropogenic emission comparisons between the a priori and *posteriori* emissions (SFBI: scale factor Bayesian inversion) for (a) CH_4 and (b) CO_2 .

Some previous studies that conducted local observations in China reported EFs in Shanxi were 6.5-7.0 m³ CH₄/ton coal (Gao et al., 2020; Zhu et al., 2017), which is about half the EF in the EDGAR v6.0 inventory. Although the comparisons in the TJM area suggested overall good agreement at the local scale in the EDGAR v6.0 inventory, our single observation site cannot represent coal mining EFs for the entire Shanxi province because of its high spatial inhomogeneity. Besides, as mentioned before that the available CH₄ EFs from underground coal mining for China (including Shanxi province) were mainly based on nationwide measurements prior to 2006 from ~800 underground mines, and were highly spatially heterogeneous (Sheng et al., 2019; Zhu et al., 2017). Besides, the observations in a few mines may not be suitable to represent averages at regional or local scales as found in this study. Furthermore, these previous EFs cannot reflect EFs changes by coal mining technology and government policy (i.e., "consolidation to large coal mines" and "phase out of small coal mines") related EFs changes in recent years (Zhang et al., 2022).

We highlight the area with low CH_4 -content coal should also be evaluated to better understand the spatial inhomogeneity of coal mining CH_4 EFs. A denser atmospheric CH_4 observation network in Shanxi province would be need to better resolve its spatial inhomogeneity for coal mining CH_4 EFs. A new CH_4 observing network is under construction and will be implemented

in our future work. Note although TROPOMI observations have not been used in CH_4 emission inversion, we also emphasize the use of satellite observations are essential in resolving the spatial inhomogeneity of greenhouse gas emissions especially for hotspots for CH_4 and CO_2 .

As mentioned above, the total anthropogenic and coal mining CH_4 emissions in China are still subject to considerable debate, not only between "top-down" atmospheric inversions and "bottom-up" inventories, but also among different inventories (i.e., EDGAR v5.0, PKU-CH₄ v2, IEA). The inversion study by Miller et al. (2019) stated that total anthropogenic CH_4 emissions increased after 2000 at rate of $1.0-1.2 \times 10^6$ tons/year and kept increasing from 2010 to 2015 at a similar rate. This increase was mainly caused by coal mining emissions. However, the comparisons between six inventories illustrated the opposite decreasing trends of coal mining emissions from 2010 to 2016. The coal mining CH_4 emissions then reversed to increasing trends that were mainly caused by increased demand for coal instead of slightly decreasing EFs. And coal production decreased from highest 4.0×10^9 tons in 2012 to the lowest 3.4×10^9 tons in 2016 by 15%, and national mean EF only deceased from 9 m³ CH₄/ton coal to 8 m³ CH₄/ton coal from 2010 to 2020 (Liu et al., 2021; Peng et al., 2016; Sheng et al., 2019). Our findings suggest that the estimation of coal mining CH₄ emissions at the regional scale, using only a single SF and without careful treatment of inhomogeneity, can potentially bring large biases.

Contrary to CH_4 , comparison for CO_2 shows that state-of-the-art inventories (EDGAR v6.0, ODIAC, PKU-CO₂) considerably underestimated anthropogenic emissions by 61%. For the SFBI method, the linear regression slope between observations and *posteriori* midday concentration was 0.82 ± 0.06 . This indicates that the underestimation in the a priori inventory has not been fully calibrated. Therefore, the true emissions are expected to be slightly

Table 2

Comparisons of Coal Mining EF for Between a Priori and Posteriori Results, Notes the a Priori Results was Derived From EDGAR v6.0 Inventory

	China (a priori)	Shanxi (a priori)	Taiyuan- Jinzhong (a priori)	Taiyuan- Jinzhong (posteriori)
CH_4 emissions (×10 ⁵ tons)	201.1	84.5	23.7	20.2(±4.3)
coal production ($\times 10^8$ tons)	35.46	8.93	1.22	1.22
EF (m ³ CH ₄ /ton coal)	7.9	13.3	26.2	23.2(±4.9)

underestimate commonly exists in all present inventories and that some large emission sources are missing in the inventories for this heavily industrialized urban area. There are a few previous studies conducted at the city scale or regional scale (i.e., in Nanjing and Nanchang), which showed less than a 20% bias in inventories (Hu et al., 2021, 2022). The extremely large SFs for the TJM area reveal large underestimation in high-density industrialized cities. A provide a study using a "bottom up" approach that approach eight inven-

higher than the SFBI result. Furthermore, the comparisons also illustrated the

previous study using a "bottom-up" approach that compared eight inventories in Beijing-Tianjin-Hebei region found the relative difference was twofold in almost all industrial cities. Further, emissions from the national database inventories (i.e., EDGAR, ODIAC) were 53%–75% lower than



provincial-level data estimated emissions (i.e., CHRED; MEIC) in the high-emitting industrial cities (Han, Zeng, Zhang, et al., 2020). Our findings provide independent support from the perspective of a "top-down" approach for this large underestimation in a heavily industrialize city in Northern China, and suggest atmospheric CO_2 concentration observations are urgently needed to monitor the true emissions from industrial cities.

4. Conclusions

To evaluate the potential CO_2 and CH_4 emission biases at the largest coal mining CH_4 emission region in China and one of the largest CO_2 emission regions, the WRF-STILT model was coupled with the most recent EDGAR v6.0 inventory to simulate the a priori concentration and then applied SFBI method to evaluate and constrain their emissions for TJM area. We found that (a) the simulated a priori CH_4 concentrations showed much better agreement with observations than CO_2 concentrations, where the regressions slope was $1.12 \pm (0.11)$ for CH_4 but $0.59 \pm (0.04)$ for CO_2 , indicating the a priori CO_2 emission was largely underestimated. (b) The derived *posteriori* anthropogenic CH_4 emissions were 92.5% of the a priori emissions, where CH_4 fugitive from coal milling accounting for 92.7% of total anthropogenic emissions. The derived CH_4 EF from coal mining was $23.2(\pm 4.9)$ m³ CH_4 /ton coal, close to the default value of high CH_4 -content coal but twofold greater than the province average, indicating large spatial inhomogeneity in coal mining CH_4 EFs for China's largest coal production area. (c) Finally, the *posteriori* CO_2 emissions were nearly 1.6-fold of the a priori emissions, illustrating some large hotspots were missing in this industrial city.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

The *posteriori* CH_4 emissions (Zhang et al., 2022) that were compared with our study can be obtained from https://doi.org/10.57760/sciencedb.02269, the EDGAR CH_4 and CO_2 emission data set (Crippa et al., 2019) can be obtained from https://edgar.jrc.ec.europa.eu/, the ODIAC CO_2 emission inventory (Oda et al., 2018) can be obtained from http://db.cger.nies.go.jp/dataset/ODIAC/DL_odiac2018.html, the MEIC inventory (Li et al., 2017) can be obtained from http://meicmodel.org/?page_id=1917, PKU inventory for CO_2 (Liu et al., 2015) can be obtained from http://inventory.pku.edu.cn/download/download.html, the Carbon Tracker 2019B emissions (Peters et al., 2007) can be obtained from https://gml.noaa.gov/aftp/products/carbontracker/co2/CT2019B/. The STILT model (Lin et al., 2003) can be downloaded from http://www.stilt-model.org/.

References

- Agustí-Panareda, A., Diamantakis, M., Massart, S., Chevallier, F., Muñoz-Sabater, J., Barré, J., et al. (2019). Modelling CO₂ weather—Why horizontal resolution matters. *Atmospheric Chemistry and Physics*, 19(11), 7347–7376. https://doi.org/10.5194/acp-19-7347-2019
- Bloom, A. A., Bowman, K. W., Lee, M., Turner, A. J., Schroeder, R., Worden, J. R., et al. (2017). A global wetland methane emissions and uncertainty dataset for atmospheric chemical transport models (WetCHARTs version 1.0). *Geoscientific Model Development*, 10(6), 2141–2156. https://doi.org/10.5194/gmd-10-2141-2017
- Brunner, D., Kuhlmann, G., Marshall, J., Clément, V., Fuhrer, O., Broquet, G., et al. (2019). Accounting for the vertical distribution of emissions in atmospheric CO₂ simulations. *Atmospheric Chemistry and Physics*, 19(7), 4541–4559. https://doi.org/10.5194/acp-19-4541-2019
- Chen, Z., Griffis, T. J., Baker, J. M., Millet, D. B., Wood, J. D., Dlugokencky, E. J., et al. (2018). Source partitioning of methane emissions and its seasonality in the U.S. Midwest. *Journal of Geophysical Research: Biogeosciences*, *123*, 646–659. https://doi.org/10.1002/2017JG004356
- Chen, Z., Jacob, D. J., Nesser, H., Sulprizio, M. P., Lorente, A., Varon, D. J., et al. (2022). Methane emissions from China: A high-resolution inversion of TROPOMI satellite observations. Atmospheric Chemistry and Physics, 22(16), 10809–10826. https://doi.org/10.5194/ acp-22-10809-2022
- Crippa, M., Guizzardi, D., Solazzo, E., Muntean, M., Schaaf, E., Monforti-Ferrario, F., et al. (2021). GHG emissions of all world countries—2021 Report, EUR 30831 EN. Publications Office of the European Union. https://doi.org/10.2760/173513
- Crippa, M., Oreggioni, G., Guizzardi, D., Muntean, M., Schaaf, E., Vullo, E. L., et al. (2019). Fossil CO₂ and GHG emissions of all world countries—2019 Report [Dataset]. Publications Office of the European Union, Luxembourg. https://doi.org/10.2760/687800
- Fang, S. X., Zhou, L. X., Tans, P. P., Ciais, P., Steinbacher, M., Xu, L., & Luan, T. (2014). In situ measurement of atmospheric CO₂ at the four WMO/GAW stations in China. Atmospheric Chemistry and Physics, 14(5), 2541–2554. https://doi.org/10.5194/acp-14-2541-2014
- Fasoli, B., Lin, J. C., Bowling, D. R., Mitchell, L., & Mendoza, D. (2018). Simulating atmospheric tracer concentrations for spatially distributed receptors: Updates to the Stochastic Time-Inverted Lagrangian Transport model's R interface (STILT-R version 2). Geoscientific Model Development, 11(7), 2813–2824. https://doi.org/10.5194/gmd-11-2813-2018
- Gao, J., Guan, C., & Zhang, B. (2020). China's CH₄ emissions from coal mining: A review of current bottom-up inventories. Science of the Total Environment, 725, 138295. https://doi.org/10.1016/j.scitotenv.2020.138295

Acknowledgments

Wei Xiao was supported by the National Key R&D Program of China (Grant 2020YFA0607501), Cheng Hu was supported by the Natural Science Foundation of Jiangsu Province (Grant BK20200802) and National Science Founding of China (Grant 42105117). Shumin Wang is supported by the Shangdianzi Background Station Open Research Project (Grant SDZ2020617) and Project on Meteorological Situation in Shanxi Province (Grant SXKMSDW20236320).



- Geels, C., Gloor, M., Ciais, P., Bousquet, P., Peylin, P., Vermeulen, A. T., et al. (2007). Comparing atmospheric transport models for future regional inversions over Europe—Part 1: Mapping the atmospheric CO₂ signals. *Atmospheric Chemistry and Physics*, 7(13), 3461–3479. https://doi.org/10.5194/acp-7-3461-2007
- Gerbig, C., Lin, J. C., Wofsy, S. C., Daube, B. C., Andrews, A. E., Stephens, B. B., et al. (2003). Toward constraining regional-scale fluxes of CO₂ with atmospheric observations over a continent: 2. Analysis of cobra data using a receptor-oriented framework. *Journal of Geophysical Research*, 108(D24), 4757. https://doi.org/10.1029/2003JD003770
- Gourdji, S. M., Yadav, V., Karion, A., Mueller, K. L., Conley, S., Ryerson, T., et al. (2018). Reducing errors in aircraft atmospheric inversion estimates of point-source emissions: The Aliso Canyon natural gas leak as a natural tracer experiment. *Environmental Research Letters*, 13(4), 045003. https://doi.org/10.1088/1748-9326/aab049
- Griffis, T. J., Chen, Z., Baker, J. M., Wood, J. D., Millet, D. B., Lee, X., et al. (2017). Nitrous oxide emissions are enhanced in a warmer and wetter world. Proceedings of the National Academy of Sciences of the United States of America, 114(45), 12081–12085. https://doi.org/10.1073/ pnas.1704552114
- Gurney, K. R., Liang, J., Roest, G., Song, Y., Mueller, K., & Lauvaux, T. (2021). Under-reporting of greenhouse gas emissions in U.S. cities. *Nature Communications*, 12(1), 553. https://doi.org/10.1038/s41467-020-20871-0
- Han, P., Zeng, N., Oda, T., Lin, X., Crippa, M., Guan, D., et al. (2020). Evaluating China's fossil-fuel CO₂ emissions from a comprehensive dataset of nine inventories. *Atmospheric Chemistry and Physics*, 20(19), 11371–11385. https://doi.org/10.5194/acp-20-11371-2020,
- Han, P., Zeng, N., Oda, T., Zhang, W., Lin, X., Liu, D., et al. (2020). A city-level comparison of fossil-fuel and industry processes-induced CO₂ emissions over the Beijing-Tianjin-Hebei region from eight emission inventories. *Carbon Balance and Management*, 15(1), 25. https://doi. org/10.1186/s13021-020-00163-2
- Hu, C., Griffis, T. J., Liu, S., Xiao, W., Hu, N., Huang, W., et al. (2019). Anthropogenic methane emission and its partitioning for the Yangtze River Delta region of China. Journal of Geophysical Research: Biogeosciences, 124, 1148–1170. https://doi.org/10.1029/2018JG004850
- Hu, C., Griffis, T. J., Xia, L., Xiao, W., Liu, C., Xiao, Q., et al. (2022). Anthropogenic CO₂ emission reduction during the COVID-19 Pandemic: Evidence from three years of urban atmospheric observations in eastern China. *Environmental Pollution*, 309, 119767. https://doi.org/10.1016/j. envpol.2022.119767
- Hu, C., Xu, J., Liu, C., Chen, Y., Yang, D., Huang, W., et al. (2021). Anthropogenic and natural controls on atmospheric δ¹³C-CO₂ variations in the Yangtze river delta: Insights from a carbon isotope modeling framework. *Atmospheric Chemistry and Physics*, 21(13), 10015–10037. https://doi.org/10.5194/acp-21-10015-2021
- Hu, L., Millet, D. B., Kim, S. Y., Wells, K. C., Griffis, T. J., Fischer, E. V., et al. (2013). North American acetone sources determined from tall tower measurements and inverse modeling. *Atmospheric Chemistry and Physics*, 13(6), 3379–3392. https://doi.org/10.5194/acp-13-3379-2013
- Janssens-Maenhout, G., Pinty, B., Dowell, M., Zunker, H., Andersson, E., Balsamo, G., et al. (2020). Toward an operational anthropogenic CO₂ emissions monitoring and verification support capacity. *Bulletin of the American Meteorological Society*, 101(8), E1439–E1451. https://doi. org/10.1175/BAMS-D-19-0017.1
- Kim, S. Y., Millet, D. B., Hu, L., Mohr, M. J., Griffis, T. J., Wen, D., et al. (2013). Constraints on carbon monoxide emissions based on tall tower measurements in the U.S. upper midwest. *Environmental Science and Technology*, 47(15), 8316–8324. https://doi.org/10.1021/es4009486
- Le Quéré, C., Peters, G. P., Friedlingstein, P., Andrew, R. M., Canadell, J. G., Davis, S. J., et al. (2021). Fossil CO₂ emissions in the post-COVID-19 era. *Nature Climate Change*, 11(3), 197–199. https://doi.org/10.1038/s41558-021-01001-0
- Li, M., Liu, H., Geng, G., Hong, C., Liu, F., Song, Y., et al. (2017). Anthropogenic emission inventories in China: A review. *National Science Review*, 4(6), 834–866. https://doi.org/10.1093/nsr/nwx150
- Lin, J. C. (2003). A near-field tool for simulating the upstream influence of atmospheric observations: The Stochastic Time-Inverted Lagrangian Transport (STILT) model. Journal of Geophysical Research, 108(D16), 4493. https://doi.org/10.1029/2002JD003161
- Lin, X., Zhang, W., Crippa, M., Peng, S., Han, P., Zeng, N., et al. (2021). A comparative study of anthropogenic CH₄ emissions over China based on the ensembles of bottom-up inventories. *Earth System Science Data*, 13(3), 1073–1088. https://doi.org/10.5194/essd-13-1073-2021
- Liu, G., Peng, S., Lin, X., Ciais, P., Li, X., Yi, X., et al. (2021). Recent slowdown of anthropogenic methane emissions in China driven by stabilized coal production. *Environmental Science and Technology Letters*, 8(9), 739–746. https://doi.org/10.1021/acs.estlett.1c00463
- Liu, Z., Guan, D., Wei, W., Davis, S. J., Ciais, P., Peng, S., et al. (2015). Reduced carbon emission estimates from fossil fuel combustion and cement production in China. *Nature*, 524(7565), 335–338. https://doi.org/10.1038/nature14677
- Lu, X., Jacob, D. J., Zhang, Y., Maasakkers, J. D., Sulprizio, M. P., Shen, L., et al. (2021). Global methane budget and trend, 2010–2017: Complementarity of inverse analyses using in situ (GLOBALVIEWplus CH₄ ObsPack) and satellite (GOSAT) observations. Atmospheric Chemistry and Physics, 21(6), 4637–4657. https://doi.org/10.5194/acp-21-4637-2021
- Maier, F., Gerbig, C., Levin, I., Super, I., Marshall, J., & Hammer, S. (2022). Effects of point source emission heights in WRF-STILT: A step towards exploiting nocturnal observations in models. *Geoscientific Model Development Discussions*, 15, 5391–5406. https://doi.org/10.5194/ gmd-2021-386
- Marcotullio, P. J., Sarzynski, A., Albrecht, J., Schulz, N., & Garcia, J. (2013). The geography of global urban greenhouse gas emissions: An exploratory analysis. *Climatic Change*, 121(4), 621–634. https://doi.org/10.1007/s10584-013-0977-z
- Miller, S. M., Matross, D. M., Andrews, A. E., Millet, D. B., Longo, M., Hirsch, A., et al. (2008). Sources of carbon monoxide and formaldehyde in North America determined from high resolution atmospheric data. *Atmospheric Chemistry and Physics*, 8(24), 7673–7696. https://doi. org/10.5194/acp-8-7673-2008
- Miller, S. M., Michalak, A. M., Detmers, R. G., Hasekamp, O. P., Bruhwiler, L. M. P., & Schwietzke, S. (2019). China's coal mine methane regulations have not curbed growing emissions. *Nature Communications*, 10(1), 303–308. https://doi.org/10.1038/s41467-018-07891-7
- Mu, L., Tian, M., Zheng, L., Li, X., & Jing, D. (2019). Characterisation and source apportionment of atmospheric organic and elemental carbon in an urban-rural fringe area of Taiyuan, China. *Environmental Chemistry*, 16(3), 187–196. https://doi.org/10.1071/en19002
- Oda, T., Maksyutov, S., & Andres, R. J. (2018). The open-source data inventory for anthropogenic CO₂, version 2016 (ODIAC2016): A global monthly fossil fuel CO₂ gridded emissions data product for tracer transport simulations and surface flux inversions. *Earth System Science Data*, 10(1), 87–107.
- Peng, S., Piao, S., Bousquet, P., Ciais, P., Li, B., Lin, X., et al. (2016). Inventory of anthropogenic methane emissions in mainland China from 1980 to 2010. Atmospheric Chemistry and Physics, 16(22), 14545–14562. https://doi.org/10.5194/acp-16-14545-2016
- Peters, W., Jacobson, A. R., Sweeney, C., Andrews, A. E., Conway, T. J., Masarie, K., et al. (2007). An atmospheric perspective on North American carbon dioxide exchange: Carbon Tracker. *Proceedings of the National Academy of Sciences of the United States of America*, 104, 18925–18930. https://doi.org/10.1073/pnas.0708986104
- Sargent, M., Barrera, Y., Nehrkorn, T., Hutyra, L. R., Gately, C. K., Jones, T., et al. (2018). Anthropogenic and biogenic CO₂ fluxes in the Boston urban region. Proceedings of the National Academy of Sciences of the United States of America, 11(29), 7491–7496. https://doi.org/10.1073/ pnas.1803715115



- Saunois, M., Bousquet, P., Poulter, B., Peregon, A., Ciais, P., Canadell, J. G., et al. (2016). The global methane budget 2000–2012. Earth System Science Data, 8, 697–751. https://doi.org/10.5194/essd-8-697-2016
- Saunois, M., Stavert, A. R., Poulter, B., Bousquet, P., Canadell, J. G., Jackson, R. B., et al. (2020). The global methane budget 2000–2017. Earth System Science Data, 12(3), 1561–1623. https://doi.org/10.5194/essd-12-1561-2020
- Scarpelli, T. R., Jacob, D. J., Maasakkers, J. D., Sulprizio, M. P., Sheng, J. X., Rose, K., et al. (2020). A global gridded (0.1° × 0.1°) inventory of methane emissions from oil, gas, and coal exploitation based on national reports to the United Nations Framework Convention on Climate Change. *Earth System Science Data*, 12(1), 563–575. https://doi.org/10.5194/essd-12-563-2020
- Seto, K. C., Hakal, S., Bigio, A., Blanco, H., Elgado, G. C., Huang, L., et al. (2014). Climate Change 2014: Mitigation of Climate Change. Contribution of Working Group III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (Chap. 12). Cambridge University Press.
- Shanxi Provincial Bureau of Statistics. (2019). Survey Office of the National Bureau of Statistics in Shanxi, Shanxi Statistical Yearbook 2018– 2019. China Statistics Press.
- Sheng, J.-X., Rachel, T., Ganesan, A. L., Maasakkers, J. D., Shen, L., Prinn, R. G., et al. (2021). Sustained methane emissions from China after 2012 despite declining coal production and rice-cultivated area. *Environmental Research Letters*, 16(10), 104018. https://doi. org/10.1088/1748-9326/ac24d1
- Sheng, J.-X., Song, S., Zhang, Y., Prinn, R., & Janssens-Maenhout, G. (2019). Bottom-up estimates of coal mine methane emissions in China: A gridded inventory, emission factors, and trends. *Environmental Science and Technology Letters*, 6(8), 473–478. https://doi.org/10.1021/acs. estlett.9b00294
- Solazzo, E., Crippa, M., Guizzardi, D., Muntean, M., Choulga, M., & Janssens-Maenhout, G. (2021). Uncertainties in the Emissions Database for Global Atmospheric Research (EDGAR) emission inventory of greenhouse gases. *Atmospheric Chemistry and Physics*, 21(7), 5655–5683. https://doi.org/10.5194/acp-21-5655-2021
- Thoning, K. W., Tans, P. P., & Komhyr, W. D. (1989). Atmospheric carbon dioxide at Mauna Loa observatory 2. Analysis of the NOAA/GMCC data, 1974–1985. Journal of Geophysical Research, 94(D6), 8549–8565. https://doi.org/10.1029/JD094iD06p08549
- Tohjima, Y., Patra, P. K., Niwa, Y., Mukai, H., Sasakawa, M., & Machida, T. (2020). Detection of fossil-fuel CO₂ plummet in China due to COVID-19 by observation at Hateruma. *Scientific Reports*, 10(1), 18688. https://doi.org/10.1038/s41598-020-75763-6
- Turnbull, J. C., Karion, A., Fischer, M. L., Faloona, I., Guilderson, T., Lehman, S. J., et al. (2011). Assessment of fossil fuel carbon dioxide and other anthropogenic trace gas emissions from airborne measurements over Sacramento, California in spring 2009. Atmospheric Chemistry and Physics, 11(2), 705–721. https://doi.org/10.5194/acp-11-705-2011
- Turnbull, J. C., Sweeney, C., Karion, A., Newberger, T., Lehman, S. J., Tans, P. P., et al. (2015). Toward quantification and source sector identification of fossil fuel CO₂, emissions from an urban area: Results from the INFLUX experiment. *Journal of Geophysical Research: Atmos*pheres, 120, 292–312. https://doi.org/10.1002/2014JD022555
- Turner, A., Kim, J., Fitzmaurice, H., Newman, C., Worthington, K., Chan, K., et al. (2020). Observed impacts of COVID-19 on urban CO₂ emissions. *Geophysical Research Letters*, 47, e2020GL090037. https://doi.org/10.1029/2020GL090037
- Turner, A. J., & Jacob, D. J. (2015). Balancing aggregation and smoothing errors in inverse models. Atmospheric Chemistry and Physics, 15(12), 7039–7048. https://doi.org/10.5194/acp-15-7039-2015
- Vardag, S. N., Gerbig, C., Janssens-Maenhout, G., & Levin, I. (2015). Estimation of continuous anthropogenic CO₂: Model-based evaluation of CO₂, CO, δ¹³C(CO₂) and Δ¹⁴C(CO₂) tracer methods. Atmospheric Chemistry and Physics, 15(22), 12705–12729. https://doi.org/10.5194/ acp-15-12705-2015
- Wang, K., Zhang, J., Cai, B., & Liang, S. (2021). Estimation of Chinese city-level anthropogenic methane emissions in 2015. Resources, Conservation and Recycling, 175, 105861. https://doi.org/10.1016/j.resconrec.2021.105861
- Wang, Y., Wang, X., Wang, K., Chevallier, F., Zhu, D., Lian, J., et al. (2022). The size of the land carbon sink in China. *Nature*, 603(7901), E7–E9. https://doi.org/10.1038/s41586-021-04255-y
- Wei, W., Li, P., Wang, H., & Song, M. (2018). Quantifying the effects of air pollution control policies: A case of Shanxi province in China. Atmospheric Pollution Research, 9(3), 429–438. https://doi.org/10.1016/j.apr.2017.11.010
- Zhang, Y., Fang, S., Chen, J., Lin, Y., Chen, Y., Liang, R., et al. (2022). Observed changes in China's methane emissions linked to policy drivers. Proceedings of the National Academy of Sciences of the United States of America, 119(41), e2202742119. https://doi.org/10.1073/ pnas.2202742119
- Zhang, Y., Gautam, R., Pandey, S., Omara, M., Maasakkers, J. D., Sadavarte, P., et al. (2020). Quantifying methane emissions from the largest oil-producing basin in the United States from space. *Science Advances*, 6(17), eaaz5120. https://doi.org/10.1126/sciadv.aaz5120
- Zhao, C., Andrews, A. E., Bianco, L., Eluszkiewicz, J., Hirsch, A., Macdonald, C., et al. (2009). Atmospheric inverse estimates of methane emissions from Central California. *Journal of Geophysical Research*, 114, D16302. https://doi.org/10.1029/2008JD011671
- Zhu, T., Bian, W., Zhang, S., Di, P., & Nie, B. (2017). An improved approach to estimate methane emissions from coal mining in China. Environmental Science & Technology, 51(21), 12072–12080. https://doi.org/10.1021/acs.est.7b01857

References From the Supporting Information

- Bares, R., Mitchell, L., Fasoli, B., Bowling, D. R., Catharine, D., Garcia, M., et al. (2019). The Utah urban carbon dioxide (UUCON) and Uintah basin greenhouse gas networks: Instrumentation, data, and measurement uncertainty. *Earth System Science Data*, 11(3), 1291–1308. https:// doi.org/10.5194/essd-11-1291-2019
- Cheng, X. L., Liu, X. M., Liu, Y. J., & Hu, F. (2018). Characteristics of CO₂ concentration and flux in the Beijing urban area. Journal of Geophysical Research: Atmospheres, 123, 1785–1801. https://doi.org/10.1002/2017JD027409
- Huang, W. J., Griffis, T. J., Hu, C., Xiao, W., & Lee, X. H. (2021). Seasonal variations of CH₄ emissions in the Yangtze River Delta region of China are driven by agricultural activities. Advances in Atmospheric Sciences, 38(9), 1537–1551. https://doi.org/10.1007/s00376-021-0383-9
- Huang, Y., Kort, E. A., Gourdji, S., Karion, A., Mueller, K., & Ware, J. (2019). Seasonally resolved excess urban methane emissions from the Baltimore/Washington, DC metropolitan region. *Environmental Science and Technology*, 53(19), 11285–11293. https://doi.org/10.1021/acs. est.9b02782
- Richardson, S., Miles, N., Davis, K. J., Lauvaux, T., Martins, D. K., Turnbull, J. C., et al. (2017). Tower measurement network of in-situ CO₂, CH₄, and CO in support of the Indianapolis FLUX (INFLUX) experiment. *Elementa: Science of the Anthropocene*, *5*, 59. https://doi.org/10.1525/elementa.140
- Saboya, E., Zazzeri, G., Graven, H., Manning, A. J., & Englund Michel, S. (2022). Continuous CH₄ and δ¹³CH₄ measurements in London demonstrate under-reported natural gas leakage. Atmospheric Chemistry and Physics, 22(5), 3595–3613. https://doi.org/10.5194/acp-22-3595-2022

- Verhulst, K. R., Karion, A., Kim, J., Salameh, P. K., Keeling, R. F., Newman, S., et al. (2017). Carbon dioxide and methane measurements from the Los Angeles Megacity Carbon Project—Part 1: Calibration, urban enhancements, and uncertainty estimates. *Atmospheric Chemistry and Physics*, 17(13), 8313–8341. https://doi.org/10.5194/acp-17-8313-2017
- Yang, Y., Zhou, M., Langerock, B., Sha, M. K., Hermans, C., Wang, T., et al. (2020). New ground-based Fourier-transform near-infrared solar absorption measurements of XCO₂, XCH₄ and XCO at Xianghe, China. *Earth System Science Data*, 12(3), 1679–1696. https://doi.org/10.5194/ essd-12-1679-2020
- Zheng, S., Wang, Y. A., & Wang, Z. Y. (2006). Methane emissions to atmosphere from coal mine in China (in Chinese, abstract in English). Safety in Coal Mines, 36, 29–33.